ORIGINAL ARTICLE



Investigating associations between nematode infection and three measures of sociality in Asian elephants

Carly L. Lynsdale^{1,2,3} · Martin W. Seltmann⁴ · Nay Oo Mon⁵ · Htoo Htoo Aung⁶ · UKyaw Nyein⁶ · Win Htut⁶ · Mirkka Lahdenperä^{7,8} · Virpi Lummaa⁴

Received: 21 February 2022 / Revised: 5 June 2022 / Accepted: 7 June 2022 $\ensuremath{\textcircled{}}$ The Author(s) 2022

Abstract

Frequent social interactions, proximity to conspecifics, and group density are main drivers of infections and parasite transmissions. However, recent theoretical and empirical studies suggest that the health benefits of sociality and group living can outweigh the costs of infection and help social individuals fight infections or increase their infection-related tolerance level. Here, we combine the advantage of studying artificially created social work groups with different demographic compositions with free-range feeding and social behaviours in semi-captive Asian elephants (*Elephas maximus*), employed in timber logging in Myanmar. We examine the link between gastro-intestinal nematode load (strongyles and *Strongyloides* spp.), estimated by faecal egg counts, and three different aspects of an elephant's social world: individual solitary behaviour, work group size, and work group sex ratio. Controlling for sex, age, origin, time since last deworming treatment, year, human sampler bias, and individual identity, we found that infection by nematodes ranged from 0 to 2720 eggs/g between and within 26 male and 45 female elephants over the 4-year study period. However, such variation was not linked to any investigated measures of sociality in either males or females. Our findings highlight the need for finer-scale studies, establishing how sociality is limited by, mitigates, or protects against infection in different ecological contexts, to fully understand the mechanisms underlying these pathways.

Significance statement

Being social involves not only benefits, such as improved health, but also costs, including increased risk of parasitism and infectious disease. We studied the relationship between and three different sociality measures—solitary behaviour, group size, and the proportion of females to males within a group—and infection by gut nematodes (roundworms), using a unique study system of semi-captive working Asian elephants. Our system allows for observing how infection is linked to social-ity measures across different social frameworks. We found that none of our social measures was associated with nematode infection in the studied elephants. Our results therefore suggest that here infection is not a large cost to group living, that it can be alleviated by the benefits of increased sociality, or that there are weak infection—sociality associations present which could not be captured and thus require finer-scale measures than those studied here. Overall, more studies are needed from a diverse range of systems that investigate specific aspects of social infection dynamics.

Keywords Host-parasite dynamics · Infection costs · Long-lived mammal · Parasite ecology · Social behaviour

Responsible Editor: T. C. M. Bakker

Carly L. Lynsdale and Martin W. Seltmann contributed equally to this work.

This article is a contribution to the Topical Collection Sociality and Disease – Guest Editors: Rebeca Rosengaus, James Traniello, and Theo Bakker.

Extended author information available on the last page of the article

Introduction

In social species, group living and social behaviours can promote reproduction and survival through numerous pathways, including increased offspring survival, increased access to potential mates and resources, protection from predation, and increased health via social support (reviewed in Cantor et al. 2021). However, the same mechanisms that offer these benefits—frequent social interactions, close proximity to conspecifics, and group density-also present costs, such as increased competition and conflict (Alexander 1974; Krause and Ruxton 2002) and risk of disease and parasite infection (McEwen 2012; Hawley et al. 2021). Different components of sociality affect parasite load in various ways. For instance, group size is positively related to intensities of non-mobile parasites, but negatively to intensities of mobile parasites (Patterson and Ruckstuhl 2013). When investigating host-parasite interactions, it is important to consider the three main components of disease, called the disease triangle: the host, the environment, and the pathogen/parasite (Scholthof 2007). Individual host characteristics such as age (Lynsdale et al. 2020) and sex (Hillegass et al. 2008), as well as behaviour and social status (Hawley et al. 2011; Keiser et al. 2016), can relate to transmission and infection risk. In addition, external factors, such as season or weather conditions, influence sickness behaviour and infection dynamics for environmentally transmitted parasites (Owen-Ashley and Wingfield 2006; Rödel and Starkloff 2014). Finally, parasites can influence host social behaviour to promote transmission of parasites from the infected to new hosts (Moore 2002; Hawley et al. 2021).

The "classic" view of social-infection dynamics is that sociality is a main driver of infections (Rifkin et al. 2012; Patterson and Ruckstuhl 2013). This view has substantial support within the literature: Higher numbers of social contacts and frequent social interaction are generally linked to increased infection (Loehle 1995; Schmid-Hempel 2017), and the reverse for increasingly solitary behaviour. A recent review on over 200 associations between individual social network measures and parasite load has shown that, within individuals, social behaviour leads to an increased risk of parasite infection (Briard and Ezenwa 2021). However, a growing body of theoretical and empirical studies challenges the assumption that infection risk and social behaviour or group size always co-vary positively (Kappeler et al. 2015; Ezenwa et al. 2016). This "enhanced" view suggests that the health benefits of sociality and group living can outweigh the costs of infection and help social individuals resist or tolerate parasites and other infectious disease (Ezenwa et al. 2016). Several studies have demonstrated that positive social interactions can be related to lower infection risk and lower infestation with gastrointestinal parasites. Social support by known group members and strong social bonds with opposite-sex partners can reduce parasite infestation (Rödel and Starkloff 2014; Müller-Klein et al. 2019), though this effect can depend on various factors such as environmental conditions or pathogen-specific transmission routes (Balasubramaniam et al. 2016). The encounter-dilution effect describes the potentially positive effects of group living with regard to costs of parasite infection (Mooring and Hart 1992), where group members experience protection from parasites by diluting the risk of being preyed on by ectoparasites or vector

species (Mooring and Hart 1992; Patterson and Ruckstuhl 2013). Additionally, social living per se can reduce the negative effects of parasite infestation and larger group size can mitigate the costs of infection with ectoparasites for group members (Almberg et al. 2015). Furthermore, the positive effects of group size have been proposed and described for endoparasites and other infectious diseases. Although reinfection with gastrointestinal nematodes is more likely for individuals in larger social groups, infected individuals benefit from larger intake of energy, which offsets main costs of nematode infection (Ezenwa and Worsley-Tonks 2018). Interestingly, the link between group size and parasite loads is often species-specific and related to other social measures. In African bovids, a positive correlation between group size and parasite infection was found, but this was only evident for relatively host-specific parasites and for hosts living in stable groups (Ezenwa 2004).

The relative infection costs versus sociality benefits of group living should be investigated under various contexts. One interesting, but understudied, factor where individual host characteristics and social group properties can interact on infection dynamics is the sex ratio of the group. Individual sex represents a dichotomy in social behaviour and immunity for many mammal species, e.g. males exhibit solitary or nomadic behaviour more often than females (Lawson Handley and Perrin 2007), which could differentially affect transmission dynamics. Life-history theory dictates that differential selection pressures, prioritising reproduction for males and longevity for females, drive sex-specific differences in resource-allocation trade-offs between immunity and reproduction (Trivers 1972; Stearns 1992; Norris and Evans 1999). Hamilton and Zuk (1982) first proposed parasitism as a mediator for these trade-offs, which can be maintained by either the boosting or regulatory effects of oestrogen and testosterone respectively on individual immune function (Folstad and Karter 1992), alongside behavioural traits that lead to differences in transmission and exposure (Patterson and Schulte-Hostedde 2011). Consequently, parasite infection intensity (i.e. parasites per host) is often higher for adult males, in comparison to females, within mammal populations (Giery and Layman 2019). Sex effects and group size effects are well investigated, but sex ratio effects on parasite infection have rarely been studied in natural systems despite the clear potential for it to influence infection dynamics within group-living species.

Asian elephants (*Elephas maximus*) are an interesting species to address those questions because they are a longlived and highly social species, with life-history traits similar to several primate and cetacean species such as humans and killer whales (*Orcinus orca*). Studies on elephants can therefore help enable generalizations across larger highly social mammals in understanding the link between sociality and parasite infection. In the wild, Asian elephants form complex social organisations, predominantly existing in matriarchal herds of matrilineal female relatives and juveniles, with males leaving the herd and becoming nomadic upon sexual maturity (Sukumar 2003). Furthermore, elephant society provides benefits such as predator defence, the transfer of social knowledge, and alloparental care of offspring (Wittemyer et al. 2005). However, elephants' high sociality also imposes costs, such as increasing costs of philopatry for older individuals and increased resource competition (Wittemyer et al. 2005), as well as facilitating the spread and persistence of parasite infections (Hawley et al. 2021). In conditions where the social setting of large long-lived mammals is artificially modulated by humans, the link between sociality and infection is less well understood. Gastro-intestinal nematodes are among the most abundant internal parasites found in Asian elephants (Fowler and Mikota 2006; Lynsdale et al. 2020), are an important driver of elephant mortality (Lynsdale et al. 2017), and are linked to reduced elephant health and immunity (Santos et al. 2020). However, the results regarding the link between parasite load and sociality in elephants are inconsistent (Vanitha et al. 2011; Abhijith et al. 2018) and warrant further investigation. Both individual host characteristics and social group properties are important predictors of parasite infection; however, the outcome of these predictions can vary depending on the classic or expanded view of the parasite-related costs and benefits of sociality (Ezenwa et al. 2016). There is a need for more empirical studies to expand the range of investigated species and systems for animals, as this will ultimately help us improve our understanding of how individuals balance the cost of parasite exposure on the one hand and the benefits of increased parasite tolerance on the other hand in the wild social living animals (Ezenwa et al. 2016).

Here, we take advantage of a unique dataset on semicaptive timber Asian elephants from Myanmar to investigate the link between sociality and parasite infection in a longlived and highly social mammal. This population is ideal for studying this relationship because their age-specific survival rates and social behaviours resemble those of wild elephants compared to those of fully captive individuals (Seltmann et al. 2018; Clubb et al. 2008; Hayward et al. 2014; Lahdenperä et al. 2016; Chapman et al. 2019; Lynch et al. 2019). In addition, the Myanma Timber Enterprise (MTE) has maintained extensive logbooks on each individual, which allow tracking individual elephants' life events, such as illness and health treatments, and provide detailed data on group compositions and friendship networks. In addition, we capitalize on the longitudinal data on parasite infection already existing in this system (Lynsdale et al. 2017, 2020), which is highly important for gaining a reliable quantification of infection dynamics as opposed to cross-sectional studies and opportunistic sampling. In these semi-captive elephants, nematode infections happen via faecal-oral horizontal transmission, which is the same route found for wild elephant populations. Adult worms live and reproduce in the gut and gastro-intestinal tract, with eggs expelled with elephant faeces (Fowler and Mikota 2006). Hence, routes of transmission of, and exposure to, local pathogens are potentially similar to those experienced by wild systems compared to fully captive systems, given the studied semi-captive individuals live in their natural habitat and express nocturnal free-roaming behaviours. Myanmar timber elephants are grouped together in mixed-sex units of approx. 4-12 individuals that work within the same forest area, overall spending at least ~ 4-8 h/day together in their working groups, for over 9 months of the year. Individuals therefore spend more time during diurnal hours within close proximity of other group members than non-group conspecifics, occupying shared physical environments where all group members can forage, defecate, and interact. This is important considering the faecal-oral environmental transmission of strongyle and Strongyloides spp. nematodes between elephant hosts (Fowler and Mikota 2006), and that Asian elephants display trunk touches around and inside other conspecifics' mouths as a form of reassurance behaviour (Plotnik and de Waal 2014). Therefore, our study system offers a unique opportunity to study the relationship between sociality and later parasite infection in a semi-experimental way, as elephants live in mixed-sex and age groups with different demographic compositions.

Our work offers data on how different measures of sociality are related to later infection in known individuals of a large, long-lived mammal, which usually roams over long distances in the wild and is therefore challenging to investigate in such detail under fully wild conditions. We use three measures of sociality and investigate their links to subsequent infection by nematode parasites. More specifically, we (1) investigate if engaging in regular social interactions with conspecifics or being solitary (individual solitary behaviour) is linked to later nematode load, measured as faecal egg counts. As wild Asian elephants exist in either strongly associated female family units, or nomadic males or loosely associated male bachelor groups, we therefore expect subsequent differences in group mitigation of infection to arise from natural sex-specific social frameworks. Specifically, we expect females to gain social benefits which protect against infection, such as elevated health and condition from increased social contact and group living, and for males to minimise infection through increased distance from, and less frequent contact with, other potential infective hosts. Transmission-related costs for adult females should be offset by benefits of higher social interaction, but not for adult males that would otherwise incur lower transmission costs from predominantly solitary lifestyles. Therefore, we predict that solitary females and social males in our study sample exhibit higher nematode loads than social females and solitary males. In addition, we (2) investigate how group size is related to nematode load. In our system, larger group size represents potential for increased nematode transmission due to higher densities of (potential) host feeding and defecating within the same habitat patches, alongside more frequent close, physical interactions, as well as improved individual health, linked to increased social interaction and social support, which may help mitigate or offset the costs of infection. We thus expected a weak but overall positive effect of group size, with individuals in larger groups yielding higher nematode load. Furthermore, we (3) studied the link between the sex ratio of the work group and nematode load. We predicted that elephants in groups that have more males than females, hence in groups with a male-biased sex ratio, show higher levels of parasite infection. In many mammals, males show higher levels of parasite infection than females (Wilson et al. 2002) and there is a need to investigate potential sex effects in the social context in which we find these elephants.

Generally, it is important to expand our understanding of the link between different social measures and parasite infection by expanding the available empirical evidence for those relationships for different species living under different conditions. This can help disentangle the complex associations of sociality and infection and the contradictory results found in previous studies, and to generate a more holistic view of a very topical problem.

Methods

Study population

The working timber elephants of Myanmar $(n \sim 3000)$ make up the largest remaining semi-captive population of this species (Mar 2007; Hedges et al. 2018). The elephants work as draught animals in logging camps during the day alongside an elephant caretaker or 'mahout', but freely roam, forage, and interact with wild and other semi-captive conspecifics at night in surrounding forest habitat (Gale 1974). The current abundance and distribution of wild elephants in Myanmar are not well studied (Leimgruber et al. 2011; Hedges et al. 2018). Myanmar's wild elephant population is estimated to be fewer than 2000 individuals (Leimgruber et al. 2011), and the chances for encounters between wild and semi-captive elephants are probably low. Though semi-captive elephants roam freely at night, they usually do not leave the wider vicinity of their timber camps. The semi-captive population is centrally managed by MTE, which mark each animal with a unique identification (ID) number on their haunches allowing for reliable recognition of different individuals. MTE staff also keep detailed records in individual log books, on e.g. elephant date of births (if captive born) or capture (if wild caught), location, maternal lineage, disease history, and treatment history, throughout an elephant's lifetime. Subsequently, MTE maintain longstanding records on Asian elephant life-history and health which are now digitised into an electronic database, allowing for accurate sampling of individuals of known age.

Trained MTE veterinarians are responsible for the basic upkeep of the elephants, and predominantly treat wounds and other working injuries. Vets are also charged with administering anthelmintic drugs (ivermectin and albendazole) approximately twice a year in accordance with state regulations as a blanket treatment rolled out across all elephants within treated camps, irrespective of their level of infection. Treatment is administered either subcutaneously (1 ml/100 kg elephant body weight), or orally (10 mg/100 kg body weight for ivermectin and 750 mg/100 kg body weight for albendazole), in line with equine guidelines. Exact dates of anthelmintic treatment are recorded onsite on the day of deworming in each animal's logbook.

The entire population is distributed across Myanmar, grouped into mixed-sex working units comprising individuals of mixed ages. Adults enter the workforce at approximately 17 years old and remain until retirement (usually around 55 years of age), with workload set by regulations on haulage ability and elephant age (Mar 2007). Elephants work only during the cold (November-February) and monsoon (June-October) seasons, and are rested in the hottest, driest months. Pregnant mothers are rested from halfway through their pregnancy (11 months), and for approximately 1-2 years following parturition where they are used for light baggage work, although calves remain nearby 'at heel' until they are weaned and can suckle as needed (Gale 1974). Following weaning and taming (at approx. age 5 years), young elephants either return to their natal group or are relocated away from their mothers. Overall, the elephants spend approx. 4-8 h/day, during diurnal hours, working and interacting with the other members of their designated groups, throughout their~40-year working life.

Sample

In total, we sampled 71 focal individuals (total no. of samples = 130 including repeated measures, no. of measures per individual = 1–6, mean = 2), all working within the Kawlin logging agency in the Sagaing Division. Our study population included 45 females (91 samples) and 26 males (39 samples), ranging in age from 10–62 years of age (mean = 26 years, median = 16 years) at the time of sampling, and of which 58 were captive born and 13 were wild caught. It was not possible to record data blind because our study involved focal animals in the field.

Sociality data collection

We investigated how infection was associated with three specific aspects of elephant sociality: individual solitary behaviour, group size, and group sex ratio. First, in order to assess an individual's direct social interactions with conspecifics (individual solitary behaviour), we extracted information from social questionnaires given to elephant handlers (mahouts) regarding whether each mahout classed their working animal as solitary (does not interact with other elephants) or social (interacts with other elephants). Mahouts can spend as long as 16 years with the same individual within this sample (Crawley et al. 2019), and thus develop an excellent knowledge of their animal and its behaviour. Ouestionnaires were carried out locally at field sites during the hot season (March - May) between 2014 and 2018. Next, using the same questionnaire data and recordings by veterinarians, we recorded the overall size of the working group of focal individuals at the time of sampling, only considering the number of adults present. Finally, we determined the sex ratio of the focal individual's work group by calculating the proportion of females in a group, excluding calves.

Faecal sampling and nematode quantification

We collected 4.5 g of fresh faecal samples (n = 130)from the 71 elephants following a standardized sampling method for our sample population (Lynsdale et al. 2015). Samples were collected within 66 days latest following social data collection, but still within the hot season of March-May. The majority of FECs were collected on the same day as the social data collection was conducted $(102/130 \text{ measures}, \sim 78\% \text{ of the total sample})$ and 1 measure was collected the following day. A further 16 FECs were collected within approximately 4–5 weeks after social data collection (25-37 days, ~12% of the total sample). Finally, 11 FECs were collected over 5 weeks after social data collection (at exactly 66 days, ~8% of the total sample). For each sample, we carried out a faecal egg count (FEC) following the special modification of the McMaster method (MAFF 1986), as in Lynsdale et al. (2020), using compound microscopes with \times 10 optical zoom and $\times 10$ magnification. We identified ova microscopically to the lowest taxonomic unit via identification of size, morphology, and developmental stage (Taylor et al. 2007; Bowman 2014). We obtained a quantified estimate of nematode load by multiplying FECs by the dilution factor (10) to convert counts into measures in eggs per gram (epg) of faeces. While FECs are a widely recognised measure of observable parasite load in veterinary and ecological studies, no study has yet provided empirical data on how FECs vary with ultimate measures of infection, e.g. intestinal worm counts, in elephants, and FECs may not account for, e.g. immature larvae, variation in shedding rates of female worms, prepatent periods, and non-reproductive individuals. As such, FECs should be regarded as a reliable estimate of the extent of infection (i.e. approximate load), rather than an exact sum of the total infective agents within a host; see Lynsdale et al. (2020) for further detail.

Statistical analysis

We analysed the association between the social landscape and subsequent infection by nematode parasites, as estimated via FECs, in our study population using three separate generalised linear mixed-effects models. All analyses were carried out in R 4.0.3 (R Core Team 2020) using glmmTMB (Anderson and Winter 2020), with untransformed FECs as the response term, and fitted to a negative binomial error structure (nbinom2), to account for the overdispersed distribution of FECs based on the mean-variance relationship of the data (Lynsdale et al. 2020). Each of our three models contained one separate univariate predictor pertaining to our sociality measures: sociality (binary, social/solitary), working group size (continuous), and working group female:male sex ratio (continuous). All models started with the same fixed covariates accounting for elephant age (continuous, years)included to the highest significant polynomial level, sex (twolevel factor, male/female), origin (two-level factor, captive born/wild caught), sample year (five-level factor, one level for each year 2014–2018), human sampler bias (three-level factor, one for each sampler who collected data), and time since last deworming treatment prior to sampling (continuous, days). However, models including year and sampler bias did not converge. Hence, we excluded sampler bias from further analyses because in models including only year and sampler bias, year was significant, whereas sampler bias was not. We also included two random factors to account for repeated measures from the same individuals (elephant ID), and from individuals located within the same working group (group ID). We tested all fixed covariate and random terms using likelihood ratio tests (LRTs), comparing starting models to replicates without each singular term in turn. Finally, as Asian elephants display clear sexual dimorphism in social structure and behaviours in the wild, we tested whether sociality-infection dynamics differed between males and females by including an interaction between sex and the social measure included (solitary behaviour, group size, group sex ratio) after excluding other non-significant covariates. Final models retained only significant confounding covariates, as well as our social terms of interest (sociality, working group size, and sex ratio). We checked models for goodness of fit using residual diagnostic checks with the DHARMa package (Hartig 2020).

Results

We found strongyle (Nematoda; Strongylidae) and Strongyloides (Nematoda; Strongyloididae) type eggs within faecal samples, observed in different developmental stages. Nematode loads varied widely between individual hosts within our population; FECs were highly skewed (aggregation parameter $\kappa = 0.272$, variance:mean ratio ≥ 1), and ranged from 0 to 2720epg (mean \pm SE = 156epg \pm 26, median = 75epg). Additionally, seven elephants (~10% of hosts, 10 faecal samples corresponding to ~8% of FEC measures) had relatively high observable levels of egg shedding with FECs over 500epg (Nielsen et al. 2010), including only one elephant having a FEC much higher than 1000epg. We did not discard this value since this is a common pattern of nematode loads found in many species. Individual time since deworming ranged from 12 to 419 days of sampling (mean = 131 days), although ~ 80%of elephants had not been dewormed within 30 days prior to sampling (n = 105/130 individuals), and 65% of elephants had not received treatment for ~ 90 days before sampling (n = 84/130). We also found variation between individuals in their social measures; however, after accounting for variance from confounding factors, the differences in FECs were not associated with those in our tested social measures (Fig. 1, Table 1). We first tested for an influence of solitary behaviour on infection rates. Overall, elephants were mostly classed as social (118/130 answers relating to 62 different elephants-84 from females, 34 from males), and we recorded only 12/130 classifications of solitary for 9 different elephants (7 answers from females and 5 from males) from the social questionnaires. Overall, males were over twice as likely to be classified as solitary, which was recorded in 19% of all males studied (n = 5/26) in comparison to 9% of the total number of females (n = 4/45). However, while mean raw FECs were 46% higher for social elephants in comparison to solitary individuals (mean raw FEC \pm SE = 164 \pm 29 epg for social elephants vs. 76 ± 22 epg for solitary conspecifics, model estimate \pm SE = 0.178 \pm 0.456), these differences were not statistically significant ($\chi^2 = 0.151$, p = 0.658). Moreover, we found no evidence for any sex-specific differences in solitary behaviour influencing infection rates when including a sex*solitary behaviour interaction (Table 2, solitary behaviour: $\chi^2 = 3.117$, p = 0.077; 164 ± 37 mean raw epg for social females vs. 164 ± 38 epg for social males, 101 ± 30 epg for solitary females vs. 40 ± 25 epg for solitary males).

We next investigated associations between FECs and the size of working groups. In our population, elephants lived in groups of varying size (range = 4–10, mean = 6.5, median = 7). After accounting for variance from treatment and from repeated measures, our results indicate that infection is lower for elephants in larger groups (model estimate \pm SE – 0.202 \pm 0.196). As with solitary behaviour, this difference was not significant ($\chi^2 = 1.044$, p = 0.307) when controlling for other contributing factors.



Fig. 1 The social landscape of infection in Asian elephants highlighting no significant variation in infection, as estimated by faecal egg counts (FEC, in eggs per gram of faeces, epg) with differences in (**a**) solitary behaviour, (**b**) working group size, and (**c**) working group sex ratio. In total, 130 measures were collected from 71 individual elephants. Red points correspond to raw FECs, black points and error

bars to mean and standard error FEC values, and black diamonds correspond to median FEC values. For (c), lines show predicted FECs, calculated in *R* using *ggpredict* (Lüdecke 2018), and shaded areas correspond to 95% confidence intervals. Plotted data is limited to FECs of 1000 epg, excluding one individual data point (2720epg)

Table 1 Effect estimates from final models for predictors of faecal egg counts for each sociality measure, fitted with or predictors	Sociality measure	Coefficient	Estimate	SE	Ζ	X^2	Р
	egg counts for each measure, fitted with ve binomial error e and log link function. group ID number was as a random effect. rcept corresponds to om elephants with nce treatment, and displayed social rather tary behaviour, (2) small working groups, ived in groups with :male sex ratio of 0. els were fitted to 130 ions from 71 elephants. ant effects ($p < 0.05$) are1. Solitary behaviour1. Solitary behaviour1. Solitary behaviour2. Group size2. Group size3. Group sex ratio3. Group sex ratio	Intercept	1.255	0.283	4.440	-	-
		Time since treatment (days)	0.007	0.001	4.528	20.427	< 0.001
structure and log link function.		Solitary behaviour (solitary)	-0.178	0.456	-0.390	0.151	0.698
Working group ID number was			Variance	SD			
included as a random effect.		Work group ID	0.800	0.894	-	-	-
The intercept corresponds to		Intercept	2.558	1.312	1.950	-	-
0 days since treatment, and		Time since treatment (days)	0.006	0.001	4.474	19.983	< 0.001
that (1) displayed social rather		Group size	-0.202	0.196	-1.030	1.044	0.307
than solitary behaviour, (2)			Variance	SD			
lived in small working groups,		Work group ID	0.762	0.873	-	-	-
a female: male sex ratio of 0.		Intercept	0.927	0.706	1.313	-	-
All models were fitted to 130		Time since treatment (days)	0.007	0.001	4.550	20.642	< 0.001
observations from 71 elephants.		Group sex ratio	0.519	1.904	0.474	0.221	0.639
Significant effects ($p < 0.05$) are in bold			Variance	SD			
		Work group ID	0.764	0.874	-	-	-

Table 2 LRT and P values for comparisons of models (as described in Table 1) but including a fixed term for sex and a social measure*sex interaction term, and replicate models consisting only of main effect terms. All models were fitted to 130 observations from 71 elephants

Sociality measure interaction	X^2	Р	
Solitary behaviour*sex	3.117	0.077	
Group size*sex	2.475	0.116	
Group sex ratio*sex	0.078	0.780	

Furthermore, we found no evidence that egg shedding in males and females differs respectively in response to more or fewer group members when testing for an interaction between FEC and group size (Table 2, $\chi^2 = 2.475$, p = 0.116).

Finally, we determined the effect of group female:male sex ratio on infection dynamics. Sex ratios of the different working groups ranged from 0.2 to 1 (mean = 0.58, median = 0.57), where an increasing ratio equals an increasing proportion of females within a group (1 = all - all female group). FECs increased with increasing sex ratio, i.e. in groups with proportionally more females (model estimate \pm SE 0.519 \pm 1.904). However, as with our other sociality measures, we found no significant association overall between working group sex ratio and later infection status in our host population ($\chi^2 = 0.221$, p = 0.639). Again, we found no evidence of sex-specific differences of group sex ratio on later infection rates when including an interaction term with sex (Table 2, $\chi^2 = 0.078$, p = 0.780). All these results are robust when data are limited to FECs and social data collected within 1 day.

Discussion

Here, we investigated associations between three specific aspects of host sociality-individual solitary behaviour, group size, and group sex ratio-and infection by gastrointestinal nematodes (strongyles and Strongyloides spp.), in a long-lived mammal with a complex social structure, the Asian elephant. Our study population consisted of 71 semicaptive Asian elephants of mixed sexes and ages, grouped into working units of various sizes (4-10 individuals) which spent diurnal hours together, but were able to display natural nocturnal roaming, socialising, and mating behaviours with other semi-captive and local wild conspecifics. We argue that this difference in social structure to their natural matriarchal herds allows for a 'natural experiment' to observe how infection is linked to the studied measures of sociality across different social frameworks. Controlling for other known confounding factors, we show that infection was not associated with any investigated measure of sociality, and that this finding was conserved across both males and females. Generally, while our results contradict our expectations, they support the argument that the parasite-related costs of sociality may vary in magnitude, are not linear, and do not operate solely in one direction. Recent studies highlight a more complex picture-that the extent of parasite-related costs, or the severity at which they are felt, may hinge on other aspects of host ecology, for example individual life history (Ezenwa et al. 2016), differences in dominance hierarchies within a social unit (Smyth and Drea 2016), or the degree of modularity or subgrouping within a population (Nunn et al. 2015). As such, there is increasing support for an 'expanded view' that infection, or the fitness costs thereof, may in fact be minimised through socially promoted resistance and/or tolerance pathways (Ezenwa et al. 2016).

While we found that males were nearly twice as likely to display solitary behaviour, neither individual solitary nor social behaviour influenced infection by strongyle or Strongyloides nematodes for any individual in our sample. When comparing across systems, infection measures are higher for social animals than those for solitary ones (Ezenwa et al. 2016). However, crucially, studies often compare the effects of infection, and selection for solitary versus gregarious behaviour, across species, rather than observing intraspecific variation within groups. Consequently, much less is known of how infection costs relate to variation in individual solitary behaviour within populations, which is an oversight considering that sociality is not homogenous within species. For example, in a number of 'social' species, individuals may realistically exhibit behaviours over a spectrum from more solitary to more social, with behavioural tendencies varying with other traits such as age and sex, as found in e.g. elephants, hamadryas baboons (Papio hamadryas hamadryas: Schreier and Swedell 2012), and Western lowland gorillas (Gorilla gorilla gorilla: Racevska and Hill 2017). Finer-scale analysis has found elephant societies show multilevel organisation and fission-fusion dynamics, with populations varying in their degree of modularity, hierarchical levels, and the extent to which these are nested (de Silva and Wittemver 2012; Nandini et al. 2018). There is also evidence that individuals maintain long-term affiliate relationships alongside ephemeral associations with conspecifics (de Silva and Wittemyer 2012), suggesting that interactiveness of both males and females to conspecifics both within and outside social units may not be temporally stable, and that tendencies of social versus solitary behaviour may change over time. Therefore, while broader classifications of sociality (i.e. as with our binary measure of solitary behaviour) are still highly valuable, especially from lesser-studied taxa, such methods may not capture the sufficient detail needed to elucidate how potential parasitism constrains finer structural contexts, for example, establishing how infection changes with increasing social contacts, or frequency or quality of interactions with other group members. While recent studies offer initial insights as to how infection operates with varying modularity in animal populations (Sah et al. 2017), there is still great scope for future studies to investigate how infection costs are incurred, and alleviated, over multi-level societies with complex coalitions, such as those seen in elephants and primates.

Group size remains one of the most widely studied predictors of parasite risk as a disease cost of sociality. Increases in group size are coupled with higher spatial-temporal concentration of potential hosts and more frequent conspecific interactions, which facilitates increased transmission risk and exposure to infective agents (Altizer et al. 2003; Rifkin et al. 2012). However, we found no evidence to support an association between nematode infection and group size. In fact, after accounting for confounding covariates, infection was overall lower for hosts in larger groups. Our results are surprising considering other studies on contagious and environmentally transmitting parasites, like gastro-intestinal nematodes, mostly show positive associations, although the size of this effect is smaller in mammalian hosts than that in birds (Rifkin et al. 2012). Instead, our results were more comparable to those on searching parasites, which are mobile enough to move between host aggregations, e.g. ectoparasites such as lice, ticks, and fleas (Rifkin et al. 2012). However, nematode motility is unlikely to substantially influence sociality-infection dynamics in our system. While elephant nematodes exhibit host-seeking behaviour in their infective stage after their third larval moult (Fowler and Mikota 2006), the distances travelled are miniscule in comparison to the roaming distances and range sizes of their elephant hosts (see Gang and Hallem 2016). While previous studies have noted a lack of association between infection and group size (Côté and Poulin 1995), it remains a relatively rare observation, possibly explained by publication bias (Rifkin et al. 2012), with group size weakly predicting parasite risk across most taxa (Côté and Poulin 1995; Rifkin et al. 2012). Interestingly, group size does not predict parasite intensity in a range of other mobile hosts (Patterson and Ruckstuhl 2013), including for other herbivore-strongyle systems, such as Grant's gazelle (Gazella granti), buffalo (Syncerus caffer), impala (Aepyceros melampus), and eland (Taurotragus oryx) (Ezenwa 2004). One possible explanation is that mobile hosts gain resistance benefits from living in larger groups, as individuals that travel over larger ranges spend less time overall within any given area and thus reduce exposure in parasite-contaminated areas (Côté and Poulin 1995). While our study elephants work within designated forest areas during the day, they are able to roam, unsupervised, over larger distances at night—a behaviour that may mitigate the parasite infection risk incurred through their diurnal work grouping.

The focus on group size as a primary predictor of infection costs is a relatively simple view of the linkage between sociality and disease, as group-living species display huge variation in both the size and structure of social landscapes. Group sex ratio, and how this factors into disease costs, is a relatively overlooked aspect of group living, which is an oversight considering how widely observed sex biases are in infection in wild systems (Wilson et al. 2002). In mammals, parasite infection intensity is often higher for adult males than females (Giery and Layman 2019)-a consequence of hormone-mediated differences in resource allocation tradeoffs between immunity and reproduction (Hamilton and Zuk 1982; Folstad and Karter 1992; Stearns 1992). Curiously, a previous work has highlighted a lack of sex-biased parasitism within the Myanmar timber elephant population, with males and females harbouring similar nematode loads across a longitudinal study period (Lynsdale et al. 2020), despite males incurring higher mortality cost of parasitism. The close proximity between individuals in mixed-sex working groups may increase transmission between males and females, possibly concealing inherent sex-specific differences in susceptibility. Yet, as with group size, infection was not associated with variation in sex ratio of working groups, suggesting that social framework does not mask differences in nematode loads between males and females in this system. Despite this, group sex ratio should be regarded as an important potential driver of associations between sociality and infection in other systems, particularly where sex biases in infection rates are observed.

The reasons underlying the absence of associations between infection and solitary behaviour therefore remain largely obscure. A possible explanation is that for reproductive-age adult elephants, nematodes are less pathogenic in comparison to e.g. bacterial and viral infections that severely impacting host survival (Fowler and Mikota 2006), or that loads do no reach critical thresholds, exacting low costs to individual morbidity and relatively weak selection pressures on sociality. However, this seems unlikely. Preliminary work has shown that observed nematode infection significantly reduces individual health and condition, as measured by white blood cell counts and liver function (Franco dos Santos D., unpublished). Moreover, our host population displays high heterogeneity in infection; nematode loads can reach exceedingly high burdens (>4000epg), beyond 'high' shedding veterinary thresholds for other non-ruminant hosts (Nielsen et al. 2013), but only for specific demographic groups (Lynsdale et al. 2020). In particular, juveniles show both elevated loads estimated via FECs (Lynsdale et al. 2020), and historically, along with adult males and nonreproductive females, suffer from increased mortality as a result of parasitism (Lynsdale et al. 2017). This, coupled with the fact that timber elephants live in mobile working units without strong competition or dominance hierarchies, may instead mean that the strongyle and Strongyloides nematodes either do not present large sociality costs in this system or that these are mitigated by the social health benefits of group living more than in other host taxa. The known variation in FECs observed across different elephant ages could provide an explanation for the lack of a link between our social measures and FECs; as acquired immune function varies across vertebrate lifespans, strong age-specific susceptibility effects may override effects of sociality in hosts, which have been previously exposed to nematodes. However, in this study, we account for age-specific variation in FECs by including host age in our analysis, allowing us to reliably detect any strong associations with sociality measures. However, it should still be noted that other individual differences in infection and exposure profiles may potentially mask weaker associations between the sociality measures investigated and FECs. Another unexplored avenue of interest is self-medicating behaviour, as observed in numerous primates (Neco et al. 2019), and in Asian elephants (Greene et al. 2020). For example, red colobus monkeys (Procolobus rufomitratus tephrosceles) increase their consumption of fodder with known anthelminthic properties, such as certain barks and Albizia spp. plants, during periods of increased shedding of whipworm (Trichuris spp.) eggs (Ghai et al. 2015). In elephants, specific plant consumption is thought to relate to self-medication behaviour for certain medical ailments, including parasitism, according to local human mahouts and knowledge holders (Greene et al. 2020). Behavioural switches to sole consumption of clay rather than vegetative matter are also noted by Asian elephants' mahouts during monsoon months, which is thought to aid in expelling established gastro-intestinal parasite infections (Greene et al. 2020). As foraging decisions can be transferred through cultural transmission in primates (Horner et al. 2006), Ezenwa et al. (2016) propose that social living and large group sizes may promote self-medicating selective foraging as a behavioural mechanism for parasite resistance. Studies have also suggested that this strategy may particularly benefit larger. longer-lived species (Neco et al. 2019), such as Asian elephants, which are also generalist browsers and graze feeders and have been known to vary their diet in response to environmental change (Sukumar 2003).

Our results provide a reliable insight into whether strong social-infection associations exist by utilizing a centralized keeping system in a rarely studied host system-semi-captive timber elephants in Myanmar. The elephant mahouts have a detailed knowledge of their elephant's behaviours and collect the elephants in the morning from the forest meaning that they are aware of whether elephants are exhibiting solitary behaviour, and whether they are grouped with the same group members or other working individuals, during part of the unsupervised period. However, it should be stressed that we cannot account for variance from any nocturnal social interactions and individual differences in foraging activity (e.g. Parker et al. 2020). While data on nocturnal activity of elephants is limited, and largely focused on fully captive systems, there is some evidence that elephants may be stationary for large periods of the night (Wilson et al. 2006; Lukacs et al. 2016), and that activity depends on age and access to outside areas (Evison et al. 2020) suggesting that most social activity takes place during diurnal hours. The measures of elephant sociality used in our study might have been too broad to capture any potential weak infection-sociality associations present in our study population or actually not capture specific social-infection mechanisms. Therefore, we cannot exclude the possibility that finer-scale measures than those investigated here might show a different picture. Data on social network dynamics and characteristics might provide the needed fine-scale measures. Specific network

components, such as connectivity or centrality within the social network, can relate to transmission dynamics (Rimbach et al. 2015). Unfortunately, the qualitative nature of our questionnaire data does not allow assessing those network characteristics in detail. Other confounding factors, such as the distribution of high-shedding individuals sharing the work areas with focal individuals and the effects of season on infection dynamics, should be noted. While faecal egg counts are moderately repeatable within hosts of our study population (Lynsdale et al. 2020), we do not know how genetic components contributed to high-shedding behaviour and hence we cannot directly control for this factor. However, including individual identity as a random factor in our analyses should help mitigate this bias to some extent. Regarding seasonal effects, this study used data collected only during one season (dry season) and hence we can exclude seasonal biases on our results, but to complete our understanding of infection-sociality associations they should also be investigated in other seasons (monsoon and cold season). Finally, although our study population shares more characteristics with wild elephant populations than fully captive populations, we suggest that our results should be treated with care when comparing to truly wild populations. The potential impact of human handling on social behaviours and group composition of our study elephants and the strong effect of regular anthelmintic treatments should be kept in mind when interpreting our results. However, some of these confounding factors constitute general challenges to studies investigating infection-sociality associations in the wild, and we were able to control for several other confounding factors of susceptibility such as age. Thus, we suggest that our findings are still a valuable addition to the literature, with very few other studies using adequate sample sizes and providing insights into the social infection dynamics of extremely long-lived terrestrial mammals.

In conclusion, our results further highlight the need for a general push towards placing social infection dynamics clearly in specific contexts, and the necessity for more studies investigating different facets of sociality from a diverse range of host-parasite systems, to inform broader meta-analyses. It is becoming increasingly clear that the relative costs of disease are determined by a number of social traits, and their organisation across different social landscapes, acting in synergy; in essence it is 'more than just a numbers game' (Nunn et al. 2015). Consequently, there is a growing emphasis on establishing how the sociality-disease nexus varies across and within a range of taxa, with elephants presenting a much-needed comparison to other long-lived, complex mammal societies. Here, we highlight the need for finer-scale studies, establishing how sociality is limited by, mitigates, or protects against infection in different ecological contexts, to fully understand the mechanisms underlying these pathways.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s00265-022-03192-8.

Acknowledgements We thank the Ministry of Natural Resources and Environmental Conservation (MONREC) in Myanmar and the Myanmar Timber Enterprise (MTE), especially the General Manager of the Extraction Department and the Managing Director of MTE, for their continued support and access to the working timber elephant logbook data and all the vets and officers involved in data collection. We especially thank Khin Than Win, Thuzar Thwin, Mumu Thein, and Ohnmar Win for logistical support in Myanmar. We thank Diogo Franco dos Santos for assistance with the collection of field measures and we thank John Jackson and Emily Lynch for discussing statistical analyses and study design. We are thankful to Simon N. Chapman for proofreading the manuscript. We are also grateful to all mahouts answering the social questionnaires. We thank two anonymous reviewers for their useful comments that helped improve the manuscript.

Author contribution CL, ML, and VL conceived and designed data collection with contributions from MWS to the development of research questions. CL, MWS, NOM, HHA, UKN, WT, and VL contributed to data collection. CL analysed the data with contributions from MWS. CL and MWS wrote the paper with input from all authors.

Funding Open Access funding provided by University of Turku (UTU) including Turku University Central Hospital. This study was financed by the Sakari Alhopuro Foundation, the KONE Foundation, the Academy of Finland, and the European Research Council.

Data and code availability Data and code are available as electronic supplementary material.

Declarations

Ethics approval The Myanmar Government and the Ethics Committee of the University of Turku approved this research. This study did not contain any procedure that would require a project license according to the Finnish National legislation (Act 497/2013 and Decree 564/2013 on the protection of animals used for scientific or educational purposes) or the EU Directive 2010/EU/63 on the protection of animals used for scientific purposes.

Competing interests The authors declare no competing interests.

Consent to participate The social questionnaires were conducted verbally in the Burmese language to aid less literate mahouts and their consent to participate was obtained.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Abhijith TV, Ashokkumar M, Dencin RT, George C (2018) Gastrointestinal parasites of Asian elephants (Elephas maximus L. 1798) in south Wayanad forest division, Kerala. India J Parasit Dis 42:382–390. https://doi.org/10.1007/s12639-018-1012-0
- Alexander RD (1974) The evolution of social organization. Annu Rev Ecol Evol S 5:325–383
- Almberg ES, Cross PC, Dobson AP, Smith DW, Metz MC, Stahler DR, Hudson PJ (2015) Social living mitigates the costs of a chronic illness in a cooperative carnivore. Ecol Lett 18:660–667. https:// doi.org/10.1111/ele.12444
- Altizer S, Nunn CL, Thrall PH et al (2003) Social organization and parasite risk in mammals: integrating theory and empirical studies. Annu Rev Ecol Evol S 34:517–547. https://doi.org/10.1146/ annurev.ecolsys.34.030102.151725
- Balasubramaniam K, Beisner B, Vandeleest J, Atwill E, McCowan B (2016) Social buffering and contact transmission: network connections have beneficial and detrimental effects on *Shigella* infection risk among captive rhesus macaques. PeerJ 4:e2630. https:// doi.org/10.7717/peerj.2630
- Bowman D (2014) Georgis' parasitology for veterinarians, 10th edn. Elsevier, St Louis, USA
- Briard L, Ezenwa VO (2021) Parasitism and host social behaviour: a meta-analysis of insights derived from social network analysis. Anim Behav 172:171–182. https://doi.org/10.1016/j.anbehav. 2020.11.010
- Cantor M, Maldonado-Chaparro AA, Beck KB et al (2021) The importance of individual-to-society feedbacks in animal ecology and evolution. J Anim Ecol 90:27–44. https://doi.org/10.1111/1365-2656.13336
- Chapman SN, Jackson J, Htut W, Lummaa V, Lahdenperä M (2019) Asian elephants exhibit post-reproductive lifespans. BMC Evol Biol 19:193. https://doi.org/10.1186/s12862-019-1513-1
- Clubb R, Rowcliffe M, Lee P, Mar KU, Moss C, Mason GJ (2008) Compromised survivorship in zoo elephants. Science 322:1649. https://doi.org/10.1126/science.1164298
- Côté IM, Poulin R (1995) Parasitism and group size in social animals: a meta-analysis. Behav Ecol 6:159–165. https://doi.org/10.1093/ beheco/6.2.159
- Crawley JAH, Lahdenperä M, Seltmann MW, Htut W, Aung HH, Nyein K, Lummaa V (2019) Investigating changes within the handling system of the largest semi-captive population of Asian elephants. PLoS ONE 14:e0209701. https://doi.org/10.1371/journal.pone. 0209701
- de Silva S, Wittemyer G (2012) A comparison of social organization in Asian elephants and African savannah elephants. Int J Primatol 33:1125–1141. https://doi.org/10.1007/s10764-011-9564-1
- dos Santos DJF, Berger V, Cristofari R, Htut W, Nyein UK, Aung HH, Reichert S, Lummaa V (2020) Seasonal variation of health in asian elephants. Conserv Physiol 8:coaa119. https://doi.org/10. 1093/conphys/coaa119
- Evison E, McKenzie A, Holmes L (2020) Social and environmental impacts on sleep in captive Asian elephants (*Elephas maximus*). Zoo Biol 39:397–404. https://doi.org/10.1002/zoo.21568
- Ezenwa VO (2004) Host social behavior and parasitic infection: a multifactorial approach. Behav Ecol 15:446–454. https://doi.org/10. 1093/beheco/arh028
- Ezenwa VO, Worsley-Tonks KEL (2018) Social living simultaneously increases infection risk and decreases the cost of infection. Proc R Soc B 285:20182142. https://doi.org/10.1098/rspb. 2018.2142
- Ezenwa VO, Ghai RR, McKay AF, Williams AE (2016) Group living and pathogen infection revisited. Curr Opin Behav Sci 12:66–72

- Folstad I, Karter AAJ (1992) Parasites, bright males, and the immunocompetence handicap. Am Nat 139:603–622. https://doi.org/ 10.1086/285346
- Fowler ME, Mikota SK (2006) Biology, medicine, and surgery of elephants. Blackwell Publishing, Ames, IA, USA
- Gale UT (1974) Burmese timber elephant. Trade Corporation, Rangoon, Burma
- Gang SS, Hallem EA (2016) Mechanisms of host seeking by parasitic nematodes. Mol Biochem Parasitol 208:23–32. https://doi.org/10. 1016/j.molbiopara.2016.05.007
- Ghai RR, Fugère V, Chapman CA, Goldberg TL, Davies TJ (2015) Sickness behaviour associated with non-lethal infections in wild primates. Proc R Soc B 282:20151436. https://doi.org/10.1098/ rspb.2015.1436
- Giery ST, Layman CA (2019) Ecological consequences of sexually selected traits: an eco-evolutionary perspective. Q Rev Biol 94:29–74. https://doi.org/10.1086/702341
- Greene AM, Panyadee P, Inta A, Huffman MA (2020) Asian elephant self-medication as a source of ethnoveterinary knowledge among Karen mahouts in northern Thailand. J Ethnopharmacol 259:112823. https://doi.org/10.1016/j.jep.2020.112823
- Hamilton WD, Zuk M (1982) Heritable true fitness and bright birds: a role for parasites? Science 218:384–387. https://doi.org/10. 1126/science.7123238
- Hawley DM, Etienne RS, Ezenwa VO, Jolles AE (2011) Does animal behavior underlie covariation between hosts' exposure to infectious agents and susceptibility to infection? Implications for disease dynamics. Integr Comp Biol 51:528–539. https://doi. org/10.1093/icb/icr062
- Hawley DM, Gibson AK, Townsend AK, Craft ME, Stephenson JF (2021) Bidirectional interactions between host social behaviour and parasites arise through ecological and evolutionary processes. Parasitology 148:274–288. https://doi.org/10.1017/ S0031182020002048
- Hayward AD, Mar KU, Lahdenperä M, Lummaa V (2014) Early reproductive investment, senescence and lifetime reproductive success in female Asian elephants. J Evol Biol 27:772–783. https://doi.org/10.1111/jeb.12350
- Hillegass MA, Waterman JM, Roth JD (2008) The influence of sex and sociality on parasite loads in an African ground squirrel. Behav Ecol 19:1006–1011. https://doi.org/10.1093/beheco/ arn070
- Horner V, Whiten A, Flynn E, de Waal FBM (2006) Faithful replication of foraging techniques along cultural transmission chains by chimpanzees and children. P Natl Acad Sci USA 103:13878–13883. https://doi.org/10.1073/pnas.0606015103
- Kappeler PM, Cremer S, Nunn CL (2015) Sociality and health: impacts of sociality on disease susceptibility and transmission in animal and human societies. Phil Trans R Soc B 370:20140116. https:// doi.org/10.1098/rstb.2014.0116
- Keiser CN, Howell KA, Pinter-Wollman N, Pruitt JN (2016) Personality composition alters the transmission of cuticular bacteria in social groups. Biol Lett 12:20160297. https://doi.org/10.1098/ rsbl.2016.0297
- Krause J, Ruxton GD (2002) Living in groups. Oxford University Press, Oxford, UK
- Lahdenperä M, Mar KU, Lummaa V (2016) Nearby grandmother enhances calf survival and reproduction in Asian elephants. Sci Rep 6:27213. https://doi.org/10.1038/srep27213
- Lawson Handley LJ, Perrin N (2007) Advances in our understanding of mammalian sex-biased dispersal. Mol Ecol 16:1559–1578. https:// doi.org/10.1111/j.1365-294X.2006.03152.x
- Leimgruber P, Min Oo Z, Aung M, Kelly DS, Wemmer C, Senior B, Songer M (2011) Current status of Asian elephants in Myanmar. Gajah 35:76–86

- Loehle C (1995) Social barriers to pathogen transmission in wild animal populations. Ecology 76:326–335. https://doi.org/10.2307/ 1941192
- Lüdecke D (2018) ggeffects: tidy data frames of marginal effects from regression models. J Open Source Softw 3:772. https://doi.org/ 10.21105/joss.00772
- Lukacs D, Poulin M, Besenthal H, Besenthal H, Fad OC, Miller SP, Atkinson JL, Finegan EJ (2016) Diurnal and nocturnal activity time budgets of Asian elephants (*Elephas maximus*) in a zoological park. Anim Behav Cogn 3:63–77. https://doi.org/10.12966/ abc.01.05.2016
- Lynch EC, Lummaa V, Htut W, Lahdenperä M (2019) Evolutionary significance of maternal kinship in a long-lived mammal. Phil Trans R Soc B 374:20180067. https://doi.org/10.1098/rstb.2018.0067
- Lynsdale CL, dos Santos DJF, Hayward AD, Mar KU, Htut W, Aung HH, Soe AT, Lummaa V (2015) A standardised faecal collection protocol for intestinal helminth egg counts in Asian elephants, *Elephas maximus*. Int J Parasitol Parasites Wildl 4:307–315. https://doi.org/10.1016/j.ijppaw.2015.06.001
- Lynsdale CL, Mumby HS, Hayward AD, Mar KU, Lummaa V (2017) Parasite-associated mortality in a long-lived mammal: variation with host age, sex, and reproduction. Ecol Evol 7:10904–10915. https://doi.org/10.1002/ece3.3559
- Lynsdale CL, Mon NO, Franco dos Santos DJ, Aung HH, Nyein UK, Htut W, Childs D, Lummaa V (2020) Demographic and reproductive associations with nematode infection in a longlived mammal. Sci Rep 10:9214. https://doi.org/10.1038/ s41598-020-66075-w
- McEwen BS (2012) Brain on stress: how the social environment gets under the skin. P Natl Acad Sci USA 109:17180–17185. https:// doi.org/10.1073/pnas.1121254109
- Moore J (2002) Parasites and the behavior of animals. Oxford University Press, Oxford, UK
- Mooring MS, Hart BL (1992) Animal grouping for protection from parasites: selfish herd and encounter-dilution effects. Behaviour 123:173–193
- Müller-Klein N, Heistermann M, Strube C, Franz M, Schülke O, Ostner J (2019) Exposure and susceptibility drive reinfection with gastrointestinal parasites in a social primate. Funct Ecol 33:1088–1098. https://doi.org/10.1111/1365-2435.13313
- Nandini S, Keerthipriya P, Vidya TNC (2018) Group size differences may mask underlying similarities in social structure: a comparison of female elephant societies. Behav Ecol 29:145–159. https://doi. org/10.1093/beheco/arx135
- Neco LC, Abelson ES, Brown A, Natterson-Horowitz B, Blumstein DT (2019) The evolution of self-medication behaviour in mammals. Biol J Linn Soc 128:373–378. https://doi.org/10.1093/bioli nnean/blz117
- Nielsen MK, Baptiste KE, Tolliver SC et al (2010) Analysis of multiyear studies in horses in Kentucky to ascertain whether counts of eggs and larvae per gram of feces are reliable indicators of numbers of strongyles and ascarids present. Vet Parasitol 174:77–84. https://doi.org/10.1016/j.vetpar.2010.08.007
- Norris K, Evans MR (1999) Ecological immunology: life history tradeoffs and immune defense in birds. Behav Ecol 11:19–26
- Nunn CL, Jordan F, McCabe CM, Verdolin JL, Fewell JH (2015) Infectious disease and group size: more than just a numbers game. Phil Trans R Soc B 370:20140111. https://doi.org/10.1098/rstb. 2014.0111
- Owen-Ashley NT, Wingfield JC (2006) Seasonal modulation of sickness behavior in free-living northwestern song sparrows (*Melo-spiza melodia morphna*). J Exp Biol 209:3062–3070. https://doi. org/10.1242/jeb.02371
- Parker J, Goldenberg S, Letitiya D, Wittemyer G (2020) Strongylid infection varies with age, sex, movement and social factors in wild

African elephants. Parasitology 147:348–359. https://doi.org/10. 1017/S0031182019001653

- Patterson JEH, Ruckstuhl KE (2013) Parasite infection and host group size: a meta-analytical review. Parasitology 140:803–813. https:// doi.org/10.1017/S0031182012002259
- Patterson LD, Schulte-Hostedde AI (2011) Behavioural correlates of parasitism and reproductive success in male eastern chipmunks, *Tamias striatus*. Anim Behav 81:1129–1137. https://doi.org/10. 1016/j.anbehav.2011.02.016
- Plotnik JM, de Waal FBM (2014) Asian elephants (*Elephas maximus*) reassure others in distress. PeerJ 2:e278. https://doi.org/10.7717/ peerj.278
- Racevska E, Hill C (2017) Personality and social dynamics of zoohoused western lowland gorillas (*Gorilla gorilla gorilla*). J Zoo Aquar Res 5:116–122. https://doi.org/10.19227/jzar.v5i3.275
- Rifkin JL, Nunn CL, Garamszegi LZ (2012) Do animals living in larger groups experience greater parasitism? A meta-analysis. Am Nat 180:70–82. https://doi.org/10.1086/666081
- Rimbach R, Bisanzio D, Galvis N, Link A, Di Fiore A, Gillespie TR (2015) Brown spider monkeys (*Ateles hybridus*): a model for differentiating the role of social networks and physical contact on parasite transmission dynamics. Phil Trans R Soc B 370:20140110. https://doi.org/10.1098/rstb.2014.0110
- Rödel HG, Starkloff A (2014) Social environment and weather during early life influence gastro-intestinal parasite loads in a groupliving mammal. Oecologia 176:389–398. https://doi.org/10.1007/ s00442-014-3017-4
- Sah P, Leu ST, Cross PC, Hudson PJ, Bansal S (2017) Unraveling the disease consequences and mechanisms of modular structure in animal social networks. P Natl Acad Sci USA 114:4165–4170. https://doi.org/10.1073/pnas.1613616114
- Schmid-Hempel P (2017) Parasites and their social hosts. Trends Parasitol 33:453–462. https://doi.org/10.1016/j.pt.2017.01.003
- Scholthof KBG (2007) The disease triangle: pathogens, the environment and society. Nat Rev Microbiol 5:152–156. https://doi.org/ 10.1038/nrmicro1596
- Schreier AL, Swedell L (2012) Ecology and sociality in a multilevel society: ecological determinants of spatial cohesion in hamadryas baboons. Am J Phys Anthropol 148:580–588. https://doi.org/10. 1002/ajpa.22076
- Seltmann MW, Helle S, Adams MJ, Mar KU, Lahdenperä M (2018) Evaluating the personality structure of semi-captive Asian elephants living in their natural habitat. R Soc Open Sci 5:172026. https://doi.org/10.1098/rsos.172026
- Smyth KN, Drea CM (2016) Patterns of parasitism in the cooperatively breeding meerkat: a cost of dominance for females. Behav Ecol 26:148–157. https://doi.org/10.1093/beheco/arv132
- Stearns SC (1992) The evolution of life histories. Oxford University Press, Oxford, UK
- Sukumar R (2003) The living elephants: evolutionary ecology, behaviour, and conservation. Oxford University Press, Oxford, UK
- Trivers R (1972) Parental investment and sexual selection. In: Campbell B (ed) Sexual selection and the descent of man. Aldine Publishing Company, Chicago, pp 136–179
- Vanitha V, Thiyagesan K, Baskaran N (2011) Prevalence of intestinal parasites among captive Asian elephants *Elephas maximus*: effect of season, host demography, and management systems in Tamil Nadu, India. J Threat Taxa 3:1527–1534. https://doi.org/10.11609/ jott.o2488.1527-34
- Wilson K, Bjornstad ON, Dobson AP, Merler S, Poglayen G, Randolph SE, Read AF, Skorping A (2002) Heterogeneities in macro-parasite infections: patterns and processes. In: Hudson PJ, Rizzoli BT, Grenfell BT, Heesterbeek H, Dobson AP (eds) The ecology of wildlife diseases. Oxford University Press, Oxford, pp 6–44

- Wilson ML, Bashaw MJ, Fountain K, Kieschnick S, Maple TL (2006) Nocturnal behavior in a group of female African elephants. Zoo Biol 25:173–186. https://doi.org/10.1002/zoo.20084
- Wittemyer G, Douglas-Hamilton I, Getz WM (2005) The socioecology of elephants: analysis of the processes creating multitiered social structures. Anim Behav 69:1357–1371. https://doi.org/10.1016/j. anbehav.2004.08.018
- Anderson EC, Winter DJ (2020) Package 'glmmTMB'. https://cran.rproject.org/web/packages/glmmTMB/index.html. Accessed Apr 2022
- Hartig F (2020) Package DHARMa: residual diagnostics for hierarchical (multi-level / mixed) regression models. https://cran.r-project. org/web/packages/DHARMa/vignettes/DHARMa.html. Accessed Apr 2022
- Hedges S, Leimgruber P, Lynam A, Mar DKU, Riddle H, Thaw UWN, Tyson M (2018) Myanmar Elephant Conservation Action Plan (MECAP): 2018–2027. Forest Department Myanmer, Yangon, Myanmar
- MAFF (1986) Manual of veterinary parasito- logical laboratory techniques. Her Majesty's Stationary Office (HMSO), London, UK

- Mar KU (2007) The demography and life history strategies of timber elephants in Myanmar. PhD thesis, University College London
- Nielsen MK, Mittel L, Grice A, Erskine M, Graves E, Vaala W, Tully RC, French DD, Bowman R, Kaplan RM (2013) American Association of Equine Practitioners: AAEP Parasite Control Guidelines, http://www.aaep.org/info/parasite-control-guidelines-231
- R Core Team (2020) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria, http://www.R-project.org
- Taylor MA, Coop RL, Wall RA (2007) Veterinary parasitology. Wiley-Blackwell, Chichester, UK

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Authors and Affiliations

Carly L. Lynsdale^{1,2,3} · Martin W. Seltmann⁴ · Nay Oo Mon⁵ · Htoo Htoo Aung⁶ · UKyaw Nyein⁶ · Win Htut⁶ · Mirkka Lahdenperä^{7,8} · Virpi Lummaa⁴

- Martin W. Seltmann martsel@utu.fi
- ¹ Natural Resources Institute, Helsinki, Finland
- ² Helsinki Institute of Life Science, University of Helsinki, Helsinki, Finland
- ³ Faculty of Biological and Environmental Sciences, University of Helsinki, Helsinki, Finland
- ⁴ Department of Biology, University of Turku, 20014 Turku, Finland

- ⁵ Department of Animal Science, University of Veterinary Science, Yezin, Myanmar
- ⁶ Myanma Timber Enterprise, Ministry of Natural Resources and Environmental Conservation, Yangon, Myanmar
- ⁷ Department of Public Health, University of Turku and Turku University Hospital, 20014 Turku, Finland
- ⁸ Centre for Population Health Research, University of Turku and Turku University Hospital, 20014 Turku, Finland